Author’s response to reviews

Title: Challenges in diagnosing dementia in patients with a migrant background - A cross-sectional study among German general practitioners

Authors:

Judith Tillmann (judith.tillmann@ukbonn.de)
Johannes Just (johannes.just@ukbonn.de)
Rieke Schnakenberg (rieke.schnakenberg@uni-oldenburg.de)
Klaus Weckbecker (klaus.weckbecker@ukbonn.de)
Birgitta Weltermann (birgitta.weltermann@ukbonn.de)
Eva Münster (eva.muenster@ukbonn.de)

Version: 2 Date: 08 Feb 2019

Author’s response to reviews:

Dear Prof. Haller,

thank you very much for this feedback. We appreciate these corrections very much. The manuscript was checked by a professional interpreter and we made further grammatical, linguistic and syntactical adjustments throughout the manuscript (marked in yellow). We particularly took into account the length of the sentence as suggested. We hope that its quality has increased as a result. We also replaced the term “influence” by “association” in the manuscript as suggested (l.42-43). We hope that it is now suitable for publication in BMC Family Practice.

Kind regards on behalf of all authors

Judith Tillmann